Author’s response to reviews

Title: Virologic suppression in response to antiretroviral therapy despite extensive resistance within HIV-1 reverse transcriptase after the first virologic failure

Authors:

Marta Iglis de Oliveira (martaiglis@gmail.com)

VALTER SOUZA JUNIOR (jr_walter@hotmail.co.uk)

Claudia Fernanda de Lacerda Vidal (vidal.claudia@gmail.com)

Paulo Sérgio Ramos de Araújo (psergiora@gmail.com)

Version: 4 Date: 17 Sep 2018

Author’s response to reviews:

We would like to thank the editors and reviewers for their comments that helped us to improve the manuscript. Herein, we present our answers to each question and changes made in the paper.

Technical Comments:
Authors email addresses

- Please include a title page at the front of your manuscript file.
  I have included it.

We appreciate the comments. Thank you.

Editor Comments:

Please check all added text carefully. For instance on page 8: The Poisson regression model was chosen... Page 8, lines 32-38: what is the 'exam' mentioned? Do teh patients take an exam? I suppose it is a mistranslation of a Portuguese word?

It was a mistranslation of a Portuguese word. I changed it

Page 8, line 40: there is twice 'associated' in the paragraph title.

I deleted 'associated'
Also check the added text on page 6, as it seems rather unclear to me.

Now is clear, I hope.

We appreciate the comments. Thank you.

Again, we appreciate all comments. We tried our best to be responsive to them. Thank you to help us improve the paper.

Best regards,

The authors.